Author's response to reviews

Title: An internet survey of 2,596 people with the fibromyalgia

Authors:

Robert M Bennett (bennetrob1@comcast.net)
Jones Jessie (jjones@Exchange.fullerton.edu)
Turk C Turk (turkdc@u.washington.edu)
Jon Russel (russell@uthscsa.edu)
Lynne Matallana (lmatallana@fmaware.org)

Version: 2 Date: 6 March 2007

Author's response to reviews:

1. The manuscript now contains the statement "The National Fibromyalgia Association received an unrestricted educational grant from Pfizer, Inc. for support of this study" under the heading "Competing Interests" placed between the Conclusions and The Authors Contributions.

2. It is noted that the reviewers did not require any revisions. However I have asterisked 2 drugs in Table 7 that are no longer available in the USA.